

# Barking up the Right Tree: Using Tree Bark to Track Airborne Particles in School Environment and Link Science to Society

A. Leite, S. Rousse, J.-f. Léon, R. Trindade, S. Haoues-Jouve, C. Carvallo, M. Dias-Alves, A. Proietti, Elise Nardin, M. Macouin

# ▶ To cite this version:

A. Leite, S. Rousse, J.-f. Léon, R. Trindade, S. Haoues-Jouve, et al.. Barking up the Right Tree: Using Tree Bark to Track Airborne Particles in School Environment and Link Science to Society. GeoHealth, 2022, 6 (9), 10.1029/2022GH000633. hal-03798486

# HAL Id: hal-03798486 https://hal.science/hal-03798486

Submitted on 5 Oct 2022

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial - NoDerivatives 4.0 International License



# GeoHealth

# **RESEARCH ARTICLE**

10.1029/2022GH000633

#### **Key Points:**

- Magnetic methods and carbon analyses evidence high concentrations of indoor and outdoor PM1 due to traffic sources
- Proximity of traffic source and absence of ventilation favorize accumulation of PM indoor threatening children's health and learning skills
- Passive monitoring using vegetal media facilitates teacher and pupil involvement and prevents eco-anxiety

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

A. d. S. Leite and M. Macouin, arualeite@gmail.com; melina.macouin@get.omp.eus

#### Citation:

Leite, A.d.S., Rousse, S., Léon, J.-F., Trindade, R. I. F., Haoues-Jouve, S., Carvallo, C., et al. (2022). Barking up the right tree: Using tree bark to track airborne particles in school environment and link science to society. *GeoHealth*, *6*, e2022GH000633. https://doi. org/10.1029/2022GH000633

Received 28 MAR 2022 Accepted 26 JUL 2022

© 2022 The Authors. GeoHealth published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

# **Barking up the Right Tree: Using Tree Bark to Track Airborne Particles in School Environment and Link Science to Society**

# A. d. S. Leite<sup>1</sup>, S. Rousse<sup>1</sup>, J.-F. Léon<sup>2</sup>, R. I. F. Trindade<sup>3</sup>, S. Haoues-Jouve<sup>4</sup>, C. Carvallo<sup>5</sup>, M. Dias-Alves<sup>2</sup>, A. Proietti<sup>6</sup>, E. Nardin<sup>1</sup>, and M. Macouin<sup>1</sup>

<sup>1</sup>Géosciences Environnement Toulouse, CNRS, IRD, Université Toulouse 3, CNES, Toulouse, France, <sup>2</sup>Laboratoire d'Aérologie, CNRS, Université Toulouse, 3, Toulouse, France, <sup>3</sup>Departamento de Geofísica, Instituto de Astronomia, Geofísica e Ciências Atmosféricas, Universidade de São Paulo, São Paulo, Brazil, <sup>4</sup>Laboratoire Interdisciplinaire Solidarités Sociétés Territoires, CNRS, Université Toulouse 2, EHESS, ENSFEA, Toulouse, France, <sup>5</sup>Institut de Minéralogie, de Physique des Matériaux et de Cosmochimie, UMR 7590, Sorbonne Université, Paris, France, <sup>6</sup>Centre de Microcaractérisation Raimond Castaing, Université Toulouse 3, Toulouse, France

**Abstract** Children's exposure to air pollution affects both their health and learning skills. Fine and ultrafine particulate matter (PM2.5, PM1), notably issued from traffic sources in urban centers, belong to the most potential harmful health hazards. However their monitoring and the society's awareness on their dangers need to be consolidated. In this study, raising teacher and pupil involvement for air quality improvement in their schools environment is reached through developing a passive monitoring technique (bio-sensors made of tree bark). The experiment was implemented in two urban elementary schools situated close to a main traffic road of the city of Toulouse (South of France). Magnetic properties, carbonaceous fraction measurements, and scanning electronic microscopy (SEM-EDX) investigations were realized both on passive bio-sensors and filters issued from active sampling. We find that traffic is the main PM1 source for both outdoors and indoors at schools. Higher levels of outdoor PM in the school's environments compared to urban background are reached especially in the cold period. The schools proximity to a main traffic source and lack of ventilation are the main causes for observed PM1 accumulation in classrooms. The co-working experiment with educational teams and pupils shows that the use of bio-sensors is a driver for children empowerment to air pollution and therefore represents a potential key tool for the teachers though limiting eco-anxiety. As PM accumulation is observed in many scholar environments across Europe, the proposed methodology is a step toward a better assessment of PM impact on pupil's health and learning skills.

**Plain Language Summary** Children's exposure to air pollution affects both their health and learning skills. Monitoring airborne particles in school environments from a knowledge co-production perspective seems essential to address air pollution in schools. Here we use tree bark and filter samples to monitor air quality in schools. Teachers and pupils are involved in the implementation process. The experiment was set up in two schools in Toulouse (France), close to the busy ring road. Traffic is the main source of air pollution both inside and outside of the classrooms. Quantities of airborne particles are higher during cold periods. The proximity of the highway and poor ventilation in the schools are the main cause of airborne particles accumulation inside classrooms. Collaboration with school staff and students shows that the use of tree has been a driving force for children to embrace the project and understand the context of air pollution, limiting anxiety related to this topic.

# 1. Introduction

Children are particularly vulnerable to air pollution exposure (Ferguson & Solo-Gabriele, 2016; Kelly & Fussell, 2015; Khreis et al., 2017; Szabados et al., 2021), one of the most important environmental causes of premature death in Europe (European Environmental Agency, 2021). They have higher air intake in comparison to adults, leading to higher absorption of pollutants through respiration, besides having immature respiratory and neurological systems (Kim et al., 2004; Martins et al., 2020; Rice & Barone, 2000). Health consequences in infants of exposure to pollutants and particulate matter (PM) may include respiratory diseases, impairments on their cardiovascular system and worsening of their neuropsychological development (WHO, 2021b and references therein). In terms of health hazards, the fine and ultrafine fraction are the most harmful ones (Martins et al., 2020; Pope & Dockery, 2006) as they may penetrate deeper into the respiratory system. The translocation from the

lungs to the bloodstream is more likely to happen in poorly soluble ultrafine particles (Pope & Dockery, 2006), thereby enhancing the deleterious effects of associated hazardous substances.

Ensuring good air quality for children's environments also provides optimal conditions for learning (Annesi-Maesano et al., 2013; Höfner & Schütze, 2021; Mendell & Heath, 2005; Vornanen-Winqvist et al., 2020), in addition to preserving their health. However, PM2.5 concentrations (i.e., particle sizes smaller than 2.5 µm) inside classrooms could be higher than outdoors, at least in European schools (Kalimeri et al., 2019). Children spend a lot of their school time indoors, making their indoor exposure to PM as important as the outdoors one. In France, students between 4 and 14 years old spend on average 3.8 hr of their daily time in schools (Zmirou et al., 2002). For a non-negligible part of the children, it is common that this time represents 10 hr on a weekday (as it includes childcare in schools). It is therefore important to assess the level of fine and ultrafine particles (PM2.5 and PM1) in air inside and outside classrooms as well as identify practices that can improve air quality.

Children's exposure depends on local outdoor air quality, indoor sources, and the ventilation system (Dimitroulopoulou, 2012). Children's lung can notably be irreversibly impacted by local traffic exposure, regardless of regional air quality (Gauderman et al., 2007). Manual ventilation of the classroom by opening the windows for example, can also induce inconveniences (temperature, wind, and ambient noise outside), which can affect the pupils' learning conditions and academic performance (Mumovic et al., 2009). Raising awareness about contaminants, sensors and the role of ventilation is not enough to create leverages toward a more active approach (Tham, 2016). Direct contextualization has been shown to be important (West et al., 2020) in environmental education. Environment education also needs to account for the risk of eco-anxiety raise (Pihkala, 2020). In the light of these facts, in this study, the approach chosen to study air pollution was to use vegetation media as bio-sensors or bioindicators (Chaparro et al., 2020; Nakazato et al., 2018; Parviainen et al., 2020). Environmental magnetism methods can be applied to natural media. They have been used to monitor anthropogenic PM trapped on vegetation such as leaves, lichens and bark (Chaparro et al., 2020; Dawai et al., 2021; Limo et al., 2018; Moreno et al., 2003; Muñoz et al., 2017; Sagnotti et al., 2009; Urbat et al., 2004; Vezzola et al., 2017; Zhang et al., 2008). Such methods rely on tracking magnetic minerals, for example, iron oxides like magnetite, hematite and maghemite (Leite et al., 2018; Marié et al., 2018; Winkler et al., 2021). For traffic-related sources, these magnetic minerals can originate from internal fuel combustion of vehicles, disk brakes abrasion and erosion of the pavement.

This study was carried out on two primary schools in Toulouse using both environmental magnetism, carbonaceous fraction measurements and scanning electronic microscopy (SEM-EDX) investigation. Our aim is to explore how the use of environmental magnetism using bio-sensors (tree bark) in addition to conventional techniques helps in transformative ways to tackle air pollution in schools by assessing the indoor/outdoor transfer of particles from traffic and by co-working with teachers and children.

# 2. Materials and Methods

#### 2.1. Location, Meteorological, and Pollutants Conditions

The city of Toulouse and its surroundings stands on quaternary alluvial terraces incised by the Garonne river. The terrace substratum is made of molasse consisting in fluvial deposits made of clay, marls lacustrine/palustrine limestones and channelized sandstones and conglomerates derived from the North Pyrenean foreland basin (Christophoul et al., 2014 and references therein).

Toulouse is the fourth biggest city in France, with a metropolitan population of approximately 1,331,000. The ring road constitutes one of the main traffic roads of Toulouse with a total of 35 km, linking important state-highways to the city traffic. It has three lanes in each way, with a mean flow of ca. 120,000 vehicles per day (in 2019) on its southern part (https://www.aua-toulouse.org/wp-content/uploads/2022/02/AUAT\_chiffres\_transports\_2022.pdf, consulted on January of 2022).

The two schools, separated by 1.3 km, are located south of downtown Toulouse, on the east bank of the Garonne river. School 1 and School 2 are located approximately 150 and 300 m north and northwest of the Toulouse ring road, respectively. Both schools' courtyards are planted with Platanus trees (Figure 1). In School 1, the total number of pupils is around 360, divided into 15 classes with an average of 24 pupils per class. This school has two courtyards, separated by a school building. The West and the East courtyards have an area of 1,400 and 3,800 m<sup>2</sup>,





**Figure 1.** (a) Satellite view of Toulouse (modified from Google Earth) with school locations and photographs of instrumented school buildings (insets). The ring road is highlighted in blue. Position of bio-sensors (yellow) and PM1 air filters location (blue) are marked with dots on the pictures. (b and c) Zoom on Schools 1 and 2 locations. SIRM<sub>M</sub> values obtained on tree barks from the sampled courtyard trees are displayed. The same color scale is used for schools 1 and 2.

respectively. School 2 counts 140 pupils for 8 classes, with classrooms of 12–26 students. It has one courtyard with an area of 4,400 m<sup>2</sup>. School buildings from School 2 and bordering West courtyard of School 1 were built in the 1950s in the same architectural-style and materials. They have classrooms of around 58 m<sup>2</sup>. Buildings from East courtyard of School 1 are historical buildings from the 1930s. In this study, courtyards and selected classrooms were sampled with bio-sensors, tree bark and air filters. In School 1, the bio-sensors and air filters were placed only in the school building bordering the West courtyard.

Concerning meteorological conditions, Toulouse has a continental climate and is located in a ventilation corridor sometimes affected by either SE wind from the Mediterranean Sea or NW winds from the Atlantic Ocean. Winters are cold and cloudy with degraded air quality due to low ventilation and emissions increase from traffic and domestic heating (https://www.atmo-occitanie.org). During spring and early summer, air quality improves due to increased ventilation and rainfall. Experiments were launched between the end of 2018 and the first half of 2019. Meteorological parameters from Blagnac Airport and air quality stations from ATMO OCCITANIE in Toulouse (Figures S1 and S2 in Supporting Information S1) show the differences between the sampled periods for NO<sub>2</sub>, PM10 and PM2.5 concentrations, daily rainfall and wind speed.

### 2.2. Methodology

Both active and passive sampling methods were deployed. The active sampling consisted in the deployment of Cyclone air filtration pumps with a cutoff size of PM1. Two different sets of samples were used for the passive sampling. The first set was composed of tree collected from the Platanus trees planted in the school courtyards, and the second one consists of 15 couples of bio-sensors made of non-contaminated Platanus tree bark squares developed in the framework of the NanoEnvi project (https://nanoenvi.omp.eu). Using both Platanus tree bark and bio-collectors derived from the co-working experiment with educational teams and pupils.

Low volume (2 L.min<sup>-1</sup>) Cyclone (BGI by Mesa Labs) with a cutoff size of 1 µm in diameter (PM1) and air filtration pumps (SKC Air Touch) were used for the active sampling. Two sampling lines were run in parallel, the first one equipped with a Whatman quartz filter and the second one with a Teflon (Polytetrafluoroethylene or PTFE filter). The lines were continuously active over one or two weeks, in January/February (period 1) and

April/June (period 2) of 2019, with six filtration lines in each season. In each school, they were deployed in one classroom hosting between 20 and 25 children (indoor, at around 2 m high) and outdoor in the courtyard (at a height of approximately 2.5 m; Figure 1). PM1 mass deposits were measured with the gravimetric method, with a high precision scale (SARTORIUS MC21S). The filters were kept in the weighing room for 24 hr at an ambient relative humidity of  $30 \pm 15\%$  before the mass measurement. Atmospheric concentrations are estimated by dividing the mass deposit by the sampled volume. Concentrations of the carbonaceous fractions were achieved through optico-thermal analyses (IMPROVE protocol allowing the separation of the carbonaceous components, Chiappini et al., 2014; Ouafo-Leumbe et al., 2018) with a DRI model 2001 thermal/optical carbon analyzer (Atmoslytic Inc., Calabasas, CA) (Chow et al., 1993, 1994, 2004, 2006). The DRI analyzer has a detection limit of 0.4, 0.1 and 0.3 µgC.cm<sup>-2</sup> for Total Carbon (TC), Elementary Carbon (EC) and Organic Carbon (OC), estimated through the instrument blank. Our measurements have an estimated accuracy of 5% for TC and 10% for EC and OC.

The capacity of bark to accumulate the airborne particles was already demonstrated for Neem trees by Dawaï et al. (2021) in Cameroon, and for various street trees by Chaprarro et al. (2020) in Buenos Aires (Argentina) and in Milan (Italy) by Vezzola et al. (2017). Here, we demonstrated that bark also efficiently accumulates PM in indoor environments as reflected by saturation isothermal remanent magnetization (SIRM) values and SEM images.

Courtyard tree barks were collected for magnetic measurements on each available tree in November and December of 2018 for School 2 and School 1, respectively. The trees are solely from the species Platanus  $\times$  accrifolia, whose bark has an annual growth cycle. A total of 48 trees were sampled (32 in School 1 and 16 in School 2). In School 1, 7 trees were sampled in the West courtyard and 25 trees in the East courtyard. In School 2, the trees sampled form two parallel lines. School 1 trees have different arrangements on each courtyard. The West one displays a single line parallel to the building, and the East one presents trees on a square disposition around the courtyard (Figure 1). Around 30 g of tree bark per tree was collected between 1 and 2 m above the ground. A small amount of each bark sample was pulverized into a fine homogeneous powder, and then put in two different gel caps sizes, with a mass of 0.1–0.8 g.

The bio-sensors consist of garlands composed of  $5/6 \sim 4 \text{ cm}^2$  squares of Platanus × acerifolia species bark pieces that are suspended from a nylon thread. The tree bark used to build the bio-sensors was collected in areas far from traffic perturbations and had SIRM measured under 1T fields to assess their low level of magnetic content. Bio-sensors were exposed in December 2018 and July 2019 (7 months), in a total of 8 rooms in School 1 and 7 rooms in School 2 (Figure 1). For each classroom, two bio-sensors were placed, indoors and outdoors, at heights of approximately 2 m to prevent children from touching them. After collection, they were ground into a fine, homogeneous powder. The powders were put into gelcaps, with masses ranging from 0.8 to 0.1 g in each gelcap.

Most of the measurements were realized at the magnetic platform at GET (University of Toulouse). Complementary analyses were performed at the paleomagnetic Laboratory at the USP (Sao Paulo, Brazil). SIRM was acquired on filters and courtyard tree bark using both a JR-6 and a JR-5 spinning magnetometer, with 2.4  $\mu$ A/m sensitivity. For the JR-6 measurements, the high sensibility mode was used, and each measurement was repeated five times. A pulse magnetizer (Magnetic Measurements Ltd.) was used to induce the SIRM, with a saturating magnetic field of 1T. The SIRM gives a qualitative concentration of the magnetic carriers (Evans & Heller, 2003). PTFE filters were chosen to carry out the magnetic investigation since in a pilot conducted in blank filters the PTFE filters had, on average, 20% the SIRM of quartz filters after 1T field induction. Filters were normalized by pumped air volume in the filters (SIRM<sub>v</sub>). Courtyard tree bark was mass normalized by sample mass (SIRM<sub>M</sub>).

Hysteresis cycles and back-field remanence curves were measured with a Vibrating Sample Magnetometer ( $\mu$ -VSM) 3900 from Princeton Measurements Corporation at the IPGP-IMPMC Mineral Magnetism Analytical Facility (Paris, France). The  $\mu$ -VSM has a sensitivity of 0.5 × 10<sup>-9</sup>Am<sup>2</sup>. Hysteresis parameters for tree bark and bio-sensors were then calculated, namely the saturation magnetization ( $M_s$ ) and coercive force ( $H_c$ ). The remanent saturation magnetization ( $M_{RS}$ ) and coercive remanent force ( $H_{CR}$ ) were obtained from the backfield remanence curve. The ratio between the hysteresis parameters ( $M_{RS}/M_S/$ and  $H_{CR}/H_C$ ) when plotted against each other creates the so-called Day Plot (Dunlop, 2002).

Bio-sensors magnetic measurement protocol consisted of alternating field (AF) demagnetization with 100 mT peak field, followed by a one-step anhysteretic remanent magnetization (ARM) acquisition and then isothermal

remanent magnetization (IRM) acquisition. The measurements were done in a 755-1.65 DC SQUID magnetometer (2G enterprises), with a sensitivity of  $10^{-12}$ Am<sup>2</sup>, located in a magnetically shielded room with an ambient field inferior to 500 nT. The ARM acquisition was performed with a superimposed AF field of 100 mT under a bias field of 30 µT. The ARM magnetization when divided by the bias field gives the susceptibility of the ARM, xARM (Evans & Heller, 2003). The IRM inductions were imparted on a pulse magnetizer (Magnetic Measurements Ltd.). All measurements were done three times and then averaged. The samples were induced in the pulse magnetizer at room temperature, for the acquisition of IRM, with field values of 1T for the SIRM and 300 mT for the backfield IRM (IRM<sub>-300mT</sub>).

The S-ratio gives the relative contribution of high and low coercivity magnetic minerals to the magnetization. The S-ratio is given by IRM<sub>-300mT</sub>/SIRM, as at 300 mT only soft coercivity magnetic carriers (magnetite, for instance) will align their magnetic moments with the applied magnetic field, and at 1T all magnetic minerals (soft and hard coercivity carriers) will align their moments with this field (King & Channell, 1991). The xARM/SIRM ratio is widely used as a proxy for grain size variation on magnetite-like magnetic minerals (Maher, 1988) because of its ability to distinguish between coarse and fine grain sizes.

Surface characteristics of the carbonaceous aerosols and of the iron oxides were obtained from representative samples of PM1 filters and bio-sensors through SEM observations using a Field-emission gun FEG SEM JEOL JSM 7100F at the Microcaracterization Center Raimond Castaing (https://centre-castaing.cnrs.fr). Composition characterization was obtained through energy dispersive X-ray spectrometer (EDS) measurements using an Oxford Instrument Detector ( $X_{Max} = 80 \text{ mm}^2$  and Ultim  $_{Max} = 100 \text{ mm}^2$ ). The samples were fixed using silver paint and coated with carbon.

# 2.3. Involvement of Pupils and Teachers in the Experiments in Schools and Social Assessment

The rollout of the experiment to the schools was conducted in several phases. After contacting the city councilor responsible for the elementary school, the first step was to establish a relationship with the school principals via the parent association (School 1) and via the local community association (School 2). The second step consisted in the visit of researchers into the school. This intervention, aiming at presenting the project to pupils and teachers, was carried out for classes of children 9–12 years old (6 classes for School 1 and 5 classes for School 2). The intervention was designed as follows: a first part (20 min) of presentation on the air quality, airborne particles and the project, a second part composed of experimentation workshops and scientific manipulations and an artistic approach, concluded in the third part by general discussion. The workshops were designed to provide an understanding of the project process by discovering the magnetic phenomena (different games and experiments with magnets and ferrofluids), the measurements by the students on a Bartington MS2 field magnetic susceptibility meter of bark samples localized on a map and by extracting magnetic particles from mud. The installation of the active pumps was the subject of a short lecture and discussion in the selected school class of each school during a third intervention. The final step involves the reporting of the results. This was only done in the two classes that hosted the pumps, as the COVID health crisis blocked possible external interventions during this time.

Teachers from the two classes that hosted the pumps were interviewed in a semi-structured manner to assess their awareness of air quality, the ventilation habits they implemented in the classrooms, and the impact of the experiments on the children and teachers.

# 3. Results

#### 3.1. Air Filters

Outdoor PM1 concentration from period 1 (January–February) ranges from 41.1 to 49.2  $\mu$ g.m<sup>-3</sup>, while outdoor PM1 concentration from period 2 (April–June) ranges from 21.1 to 49.8  $\mu$ g.m<sup>-3</sup> (Table 1). Indoors, concentration values reveal higher levels during period 1 (varying from 35.7 to 40.7  $\mu$ g.m<sup>-3</sup> for School 2 and 1, respectively) than during period 2 (ranging from 24.2 to 29.3  $\mu$ g.m<sup>-3</sup> for School 2 and 1, respectively). During the first half of the year 2019, the ambient daily mean concentrations of NO<sub>2</sub> and PM2.5 recorded by AtmoOccitanie at the traffic station (route d'Albi—about 8 km to the North of the schools) range between 8.0 and 80  $\mu$ g.m<sup>-3</sup> and 2.0 and 44  $\mu$ g.m<sup>-3</sup>, respectively (Figure 2). This rather large daily variability reflects the proximity of traffic emissions.

# Table 1

Elemental (EC) and Organic Carbon (OC), PM1 Concentrations, and SIRM Measured at School #1 and #2 on PTFE Filters During the 4 Field Experiments in 2019

Date		EC (µgm <sup>-3</sup> )	OC (µgm <sup>-3</sup> )	OC/EC	PM1 (µgm <sup>-3</sup> )	SIRM (10 <sup>-10</sup> Am <sup>2</sup> )
School #1						
14–26 January	Indoor	0.7	4.7	6.7	40.7	2.87
	Outdoor	1.6	4.4	2.7	49.2	3.21
	I/O (no unit)	0.4	1.1	-	0.8	0.9
08–18 April	Indoor	0.8	5.5	6.9	29.2	0.97
	Outdoor	0.9	2.9	3.2	21.1	0.49
	I/O (no unit)	0.9	1.9	-	1.4	1.9
School #2						
28 January–09 February	Indoor	0.4	4.9	12.2	35.7	1.08
	Outdoor	1.4	3.1	2.1	41.1	1.98
	I/O (no unit)	0.3	1.6	-	0.9	0.5
17–27 June	Indoor	0.7	2.8	4.0	24.3	0.42
	Outdoor	0.8	2.6	3.2	49.8	1.94
	I/O (no unit)	0.9	1.1	-	0.5	0.2

Note. The ratio between indoor and outdoor concentration (I/O) is indicated for each parameter and sampling period in 2019.

The average PM2.5 concentration during the sampling periods of the filters (see Table 1) is on average 3 times lower than the PM1 filter concentration.

Figure 2 presents the measured PM1 concentration on the school filters alongside the daily measured PM2.5 and  $NO_2$  concentration at official air quality stations (ATMO OCCITANIE). The station is located in Toulouse (route Albi) and represents the traffic background of the city. Generally, the school filters display higher PM1 concentrations during period 1 compared to the PM2.5 measurements from the air quality stations. During period 2, the PM1 concentration on the school filters is comparable to the highest PM2.5 concentration values from the air quality stations. This shows an overall higher concentration of PM in the school environments compared to the urban background of Toulouse. The  $NO_2$  concentration patterns follow the PM1 concentration patterns measured in both schools.

The elemental carbon (EC) indoor concentrations reach a maximum value  $(0.8 \ \mu g.m^{-3})$  in School 1 during period 2. The minimum concentration is reached in School 2 during period 1  $(0.4 \ \mu g.m^{-3})$ . In School 1, the values are





Table 2

SIRM<sub>M</sub>, S-Ratio and xARM/SIRM Means and Standard Deviation Measured on Bio-Sensor at Schools 1 and 2

1117							
School #1		SIRM (10 <sup>-5</sup> Am <sup>2</sup> .kg <sup>-1</sup> )	S-ratio	xARM/SIRM (10 <sup>-4</sup> m.A <sup>-1</sup> )	Ν		
Garland bio-sensors 13/12/2018 04/07/2019 I/O (no unit)	Indoor	$1.35 \pm 0.46$	$0.95 \pm 0.01$	$9.14 \pm 3.04$	8		
	Outdoor	$1.88 \pm 0.49$	$0.97 \pm 0.02$	$4.83 \pm 2.76$	8		
		0.7	-	-	-		
Courtyard tree bark	West	$6.40 \pm 3.50$	$0.89 \pm 0.07$	-	7		
	East	$10.62 \pm 3.26$	$0.92 \pm 0.04$	-	25		
School #2							
Garland bio-sensors 20/12/2018-28/06/2019 I/O (no unit)	Indoor	$0.95 \pm 0.21$	$0.93 \pm 0.05$	$5.69 \pm 3.27$	7		
	Outdoor	$1.08 \pm 0.43$	$0.89 \pm 0.05$	$5.31 \pm 3.88$	6		
		0.9	-	-	-		
Courtyard tree bark		8.61 ± 3.71	$0.87 \pm 0.05$	_	18		

Note. N is the number of samples and I/O is the ratio between indoor and outdoor SIRM<sub>M</sub>.

almost stable throughout the experiments. In school 2, the value almost doubles from period 1 to 2. Outdoor EC concentrations are higher than indoor, and higher during period 1, reaching a maximum in School 1 ( $1.6 \mu g.m^{-3}$ ). The minimum value is recorded for the outdoor environment in School 2 during period 2 ( $0.8 \mu g.m^{-3}$ ).

The values of organic carbon (OC) concentration for the filters are higher indoors (Table 1), with a maximum value in School 1 of 5.5  $\mu$ g.m<sup>-3</sup> during period 2 and a minimum value in School 2 of 2.8  $\mu$ g.m<sup>-3</sup> in the same period. The outdoor values are maximum in School 1 during period 1 (4.4  $\mu$ g.m<sup>-3</sup>) and minimum in School 2 during period 2 (2.6  $\mu$ g.m<sup>-3</sup>).

The OC/EC ratio can be used to infer the source of the carbonaceous aerosols and also the formation of secondary organic carbon (SOC, Xu et al., 2015). The values observed in the schools are higher indoor than outdoor (Table 1). The indoor values have a small variation between the periods in School 1, with a maximum of 6.9 during period 2. In School 2, the OC/EC ratio decreases from 12.2 to 4.0 from periods 1 to 2. The outdoor ratio values have less variation, having a maximum in period 2 of 3.2 in both schools and a minimum at School 2 of 2.1 for period 1.

SIRM<sub>v</sub> measurements were carried out on the Teflon filters. It allows direct comparison between the concentration of iron oxides captured by the filters and PM1 concentration measured on the Quartz filters. SIRM<sub>v</sub> values in School 1 have maximum and minimum measured values of  $3.21 \times 10^{-10}$  and  $0.49 \times 10^{-10}$ A.m<sup>-1</sup> in the outdoor environment in periods 1 and 2, respectively. In School 2, maximum and minimum values are reported outdoors during period 1 and indoors during period 2, with measured values of  $1.98 \times 10^{-10}$  and  $0.42 \times 10^{-10}$ A.m<sup>-1</sup>, respectively.

### 3.2. Courtyard Tree Bark and Bio-Sensors

SIRM<sub>M</sub> values for the courtyard tree bark samples vary between  $17.68 \times 10^{-5} \text{ Am}^2 \text{.kg}^{-1}$  and  $3.63 \times 10^{-5} \text{ Am}^2$ . kg<sup>-1</sup> (Figure 1). Mean SIRM<sub>M</sub> values are close in all courtyards, ranging from  $6.40(\pm 3.50) \times 10^{-5} \text{Am}^2 \text{.kg}^{-1}$  (on School 1 West courtyard) to  $10.62(\pm 3.26) \times 10^{-5} \text{ Am}^2 \text{.kg}^{-1}$  (on School 1 East courtyard). In School 2, tree bark SIRM values have a mean of  $8.61(\pm 3.71) \times 10^{-5} \text{ Am}^2 \text{.kg}^{-1}$ . In School 1, a difference in the distribution of SIRM<sub>M</sub> values is noticeable between both courtyards, with higher values in the Eastern courtyard.

Bio-sensors were placed inside and outside the classrooms, on different floors of the building. SIRM<sub>M</sub> values for both indoor and outdoor bio-sensors of School 1 are higher than those obtained from School 2 (Table 2). In School 1, indoor bio-sensors have mean SIRM<sub>M</sub> values of  $1.35(\pm0.46) \times 10^{-5}$  Am<sup>2</sup>.kg<sup>-1</sup> - whereas mean values for outdoor samples are  $1.88(\pm0.49) \times 10^{-5}$  Am<sup>2</sup>.kg<sup>-1</sup>. SIRM<sub>M</sub> mean values for School 2 bio-sensors are  $0.95(\pm0.23) \times 10^{-5}$  Am<sup>2</sup>.kg<sup>-1</sup> (indoor) and  $1.08(\pm0.43) \times 10^{-5}$  Am<sup>2</sup>.kg<sup>-1</sup> (outdoor). The SIRM<sub>M</sub> mean values for bio-sensors display a similar trend than SIRM<sub>V</sub> and EC concentrations measured in the air filters, with higher values outdoors, in School 1.



**Figure 3.** Day plot of the tree bark (circles) and previously published data (crosses). Data from Q. ilex leaves are from Szöyi et al. (2007), data for PM10 filters from Sagnotti et al. (2006), data for gasoline, brakes and diesel from Sagnotti et al. (2009) and Sagnotti and Winkler (2012), and data from dust (heated and non-heated) are from Górka-Kostrubiec et al. (2019).

The S-ratios calculated for the courtyards' barks display similar values for the three yards with  $0.92(\pm 0.04)$ ,  $0.89(\pm 0.07)$  and  $0.87(\pm 0.05)$  for School 1 East and West ones and School 2 one, respectively. Results of the bio-sensors (Table 2) show a greater variability on School 2 when compared to School 1. Mean S-Ratio calculated for bio-sensors of School 1 are  $0.95 (\pm 0.01)$  and  $0.97 (\pm 0.02)$  in indoor and outdoor environments respectively. School 2 presents lower mean S-ratio values in comparison with the bio-sensors from School 1, with indoor and outdoor values at  $0.93(\pm 0.05)$  and  $0.89(\pm 0.05)$  respectively, indicating a higher content of high coercivity minerals than for School 1 samples.

Hysteresis parameters from both tree bark from the schoolyards' trunks and bark used as bio-sensors displayed a large range of parameters (Figure 3) compared, for example, to data obtained in Roma (Italy) on leaves (Szöyi et al., 2007) and on PM10 filters (Sagnotti et al., 2006). They extend to values obtained by Sagnotti et al. (2009) and Sagnotti and Winkler (2012) from exhaust residues from gasoline and diesel motors as well from brake wear dust collected directly in situ. This could result from different sources of particles, meaning different magnetic mineralogy such as high coercivity minerals like hematite(of geological origin e.g.,) or metallic iron (linked to abrasion of combustion cylinder, pads and disk brakes), which influences SIRM and coercivity values (Gorka-Kostrubiec & Szczepaniak, 2017; Gorka-Kostrubiec et al., 2019) and therefore impact the S-Ratio and/ or grain sizes parameters.

The highest xARM/SIRM values are found at the indoor environment of School 1 (Figure 4, Table 2), with a mean of  $9.14(\pm 3.04) \times 10^{-4}$ m.A<sup>-1</sup>, whereas the outdoor ones have a mean of  $4.83(\pm 2.76) \times 10^{-4}$ m.A<sup>-1</sup>.







School 2 displays similar calculated mean values in the indoor and outdoor environments, with values of  $5.69(\pm 3.27) \times 10^{-4}$ m.A<sup>-1</sup> and  $5.31(\pm 3.88) \times 10^{-4}$ m.A<sup>-1</sup>, respectively. School 2 mean values are also similar to the outdoor value of School 1.

The ratio of the indoor to the outdoor parameters (hereinafter called I/O; Table 2) is a simple and straightforward way to quantify the indoor-outdoor concentration relationship of a given compound, being used widely in air quality studies (Kalimeri et al., 2019 and references therein). I/O for EC shows similar behavior in both schools, increasing from period 1 to 2, with values of 0.4–0.8, and 0.3–0.9 for School 1 and 2, respectively (Table 2). Similarly, I/O for PM1 shows an increasing trend toward summer from 0.8 to 1.4 and 0.9 to 1.3, in School 1 and 2 respectively. I/O for SIRM<sub>V</sub> goes from 0.9 to 2.0 from period 1 to 2, showing more than a two-fold increase, whereas in School 2 there is a decrease from 0.5 to 0.2. The I/O for SIRM<sub>M</sub> on both school bio-sensors have close values.

#### 3.3. SEM Observations

SEM results on the filters show the dense quartz matrix of the filters, encrusted with agglomerations of carbon "fluffy" soot aggregates, iron oxides aggregates and spherules. Figures 5a and 5b show backscattered electron images (BSE) from School 1 outdoor and indoor filters respectively (collected in period 2). For the outdoor filter (Figure 5a) one can notice an agglomeration containing iron (particle number 1), aluminum, and sulfur, with a dimension of 2.1  $\mu$ m, and an elongated shape. Some carbon-rich agglomerates are also visible, for instance particle 2, 1  $\mu$ m in size, mainly composed of carbon with some Na. Particle 3 is a 82 nm in diameter spherule, mainly composed of iron and oxygen. A secondary electron image (SE) with greater magnification (X40,000) (Figure 5a) enables further investigation of the topography of the iron oxide spherule (particle 3) and highlights the spherule and soot aggregate cementation to the quartz fiber.

The SEM image of the indoor filter (Figure 5b) shows an agglomeration of iron oxide spherules, attached within a carbon "fluffy" soot aggregate (dimension of 1  $\mu$ m) on the quartz fiber. There are at least 6 spherules trapped (agglomerate 1), with sizes of 40, 57, 99, 123, 128 and 157 nm. Secondary electron image (Figure 5b) displays in greater detail the spherule agglomeration, cemented on the quartz filter fiber with soot aggregates.





**Figure 5.** A: Backscattered SEM image of School 1 outdoor filter iron from period 2 with spectra of oxide spherule, iron oxide agglomerations and soot aggregates; (a) Magnified part of the image in Secondary electron image centered on the iron oxide spherule (particle 3). B: Backscattered SEM image of School 1 indoor filter from period 2 showing agglomeration of at least six iron oxides spherules (1) and carbon-rich aggregates (2); (b) Secondary electron image of agglomeration of iron oxide spherules (1); C: Backscatter SEM image from outdoor bio-sensors exposed in School 1 showing several agglomerated-like particles with irregular shapes including iron oxides.

Figure 5c displays typical particles from an outdoor bio-sensor in School 1. Particles are arranged in irregularly shaped agglomerates. Particle 1 is an iron oxide, with a dimension of 1.3  $\mu$ m. Spectrum from particle 2 has a classical lithic composition (containing Mg, Al, Si, K and Na, Al and Si respectively) and size of 1.3  $\mu$ m.

# 3.4. Teacher's Interviews

Two teachers were interviewed. One of the teachers reported that there are no official guidelines regarding air pollution except for avoiding outdoor sports during pollution peaks. Nevertheless, a poor air quality in the school environments linked to traffic emission is assumed by both teachers due to the visible proximity with the main traffic road for School 1 and an important access road to the city for School 2.

For the latter, occasional trash burning nearby the school is also mentioned. One of the teachers has a scientific background and says that his classrooms are receptive to topics concerning the environment. This teacher was involved in the design of the intervention by asking the researchers to come and explain the needs of children. From the teacher's point of view and observations, the fact that the researchers came to show the sensors made from bark, which is an inexpensive setup, was key in the appropriation of the project by the children. For this teacher, even if children didn't understand all the scientific background behind it, this is not problematic. What both teachers revealed is the importance that children were able to meet the researchers on more than one occasion, and to touch and to use the experimental material in their classrooms. Both teachers reported the problem of noise when the windows are open. Noises arise from children's activities in the schoolyard, helicopters and traffic, notably the honking of car and vehicle noises that reflect anger and stress. Besides the noise and difficulty of opening windows due to their dilapidated state, one of the teachers also reported some reluctance to vent in order to prevent traffic pollution from entering the classroom. The classroom is ventilated by the windows in the morning during cleaning before the arrival of children and sometimes during the day. Managing troubles in concentration and ventilating with an appropriate frequency is difficult for the teacher. The opening of windows during the day is induced by human sweat smelling and heating during the spring months.

Inconvenience due to the noise generated by the cyclone pump inside the classroom was reported by the pupils to their teachers after a few days of the experiment running. The teacher also reported relief at the end of the experiment with the active system despite the great motivation of pupils and teachers. Therefore, noise related to common active systems may prevent teachers' involvement in monitoring air quality in their classrooms while developing passive vegetal-oriented captors systems might be a way to develop their collaboration.

# 4. Discussion

### 4.1. Outdoor School Environment

The data indicates that the main source of the PM1 captured in both schools by the air filters is traffic emissions. Traffic emissions are generated by internal engine combustion and abrasion from the vehicle's parts (disk brakes for instance) and asphalt. Outdoors, Viana et al. (2007) reports OC/EC ratio values characteristic of traffic emissions in urban settings during winter and summer for Belgium (4.4 and 3.5, respectively), The Netherlands (4.7 and 2.8, respectively) and Spain (5.6 and 4.6, respectively). In Athens, OC/EC ratios of 4.7 and 2.9 (Paraskevopoulou et al., 2014; Remoundaki et al., 2013, respectively) were found on traffic-related urban sites. Reported OC/EC ratios from outdoor samples in this study have a similar range compared with previous OC/EC ratios in European cities, pointing out traffic emission (specifically internal diesel combustion) as the main source of the captured PM1 in this study.

This traffic-sourced origin is further reinforced by the similarity of the measured trends obtained from the daily variations in NO, and PM2.5 concentrations (Figure 2). NO, is a secondary pollutant formed mainly from nitric oxide (NO), the predominant species in vehicle exhaust (Henderson et al., 2007). The covariation found between PM2.5 and NO<sub>2</sub> indicates that vehicle exhaust is the main source of PM2.5 in the city of Toulouse. The presence of iron-rich fly ash spherules among soot carbon chains (Figures 5a and 5b), on the PM1 filters, is also an indicator of internal combustion engine emissions found both indoors and outdoors (Leite et al., 2021; Shi et al., 2015). The presence of magnetite in PM is often linked to combustion and traffic sources, in line with the shapes detected for some iron oxides on the PM 1 filters (Figures 5a and 5b) by the SEM microscopy (spherules—Bardeli et al., 2011; Castañeda-Miranda et al., 2014; Gonet & Maher, 2019; Gonet et al., 2021; Leite et al., 2021; Liati et al., 2015; Sagnotti & Winkler, 2012; Winkler et al., 2021). The magnetic mineralogy of airborne iron oxides captured on tree bark and bio-sensors is composed of low coercivity carriers (magnetite-like) and high coercivity magnetic grains (e.g., hematite, goethite) as suggested by the S-ratio values between 0.89 and 0.97 (outdoors of Schools 2 and 1 respectively; Bloemendal et al., 1992; Frank & Nowaczyk, 2008). Some of the magnetic grains present in PM may have a geological origin, for instance being carried by wind over long distances (Formenti et al., 2014). Nevertheless, the magnetization used here (SIRM and ARM) are mostly dominated by the magnetite-like component.

Since traffic is the main source for PM in these school environments, the variability in SIRMs between courtyards, interpreted as variation of PM accumulation, could be caused by the impact of building configurations/ architectural features (Figure 1). School 1 (West) and School 2 courtyards have similar configurations and similar SIRM values. Both present a tree line parallel to the main 3-floors school building making a corridor shape with the adjacent building that may increase venting. Conversely, School 1 (East) courtyard presents a u-shape configuration with the open side facing a sports ground and the ring road. The higher SIRM values observed in this courtyard could result from either the highest influence of the ring road (due to its proximity and open setting, Hofman & Samson, 2014; Hofman et al., 2013) and/or from the enclaved configuration limiting the venting.

Measured PM1 values in the schoolyards revealed a higher content in PM in both school environments than the background PM concentration from Toulouse monitored by official stations and especially during Period 1. This shows the influence from traffic emission sources in the studied schools due to their proximity to important major highways and access streets. Moreover, the two schools from Toulouse present comparatively high concentrations of PM1 compared to PM1 and PM2.5 data from European schools located in big urban centers (Figure 6). Outdoors, concentrations recorded in Toulouse are comparable to Western European cities such as Milan, Munich and Barcelona, all of which have larger populations (1.4 M, 1.5 M and 1.6 M respectively) than Toulouse (0.5 M). Each school has a particular context in a given city, explaining the large scatter in the recorded values (Figure 6).

#### 4.2. School's Indoor Environments

As there are no indoor combustion sources, EC is a marker of outdoor air penetration. The I/O for EC increases between the first and the second period, indicating a higher influence of outdoor air in the classrooms with the changing of the seasons and temperature, due to more frequent opening of the windows as reported by the teachers. The mean I/O for EC for both sampled season periods equals 0.6 in both schools, similar to the one obtained from total magnetic PM measured on bio-sensors (0.6 and 0.9, respectively for School 1 and 2). The magnetic fraction of PM found indoors also points to external sources, hence traffic sources, in agreement with previous studies that found that magnetic particles on household dust are derived from traffic and industrial activities (Beata Górka-Kostrubiec & Iga Szczepaniak-Wnuk, 2017; Jelenska et al., 2017; Kelepertzis et al., 2019). Other known potential sources of magnetic PM indoors are cooking, combustion processes, smoking (Jordanova et al., 2006) and laser printing (ink composed of nanoparticles, Pirela et al., 2015), all of which are absent in the studied schools.





Figure 6. PM1 and PM2.5 mean concentrations in primary schools across Europe, in indoor and outdoor environments. See references from the supporting information.

High indoor values of OC concentration (Table 1), reflected on I/O for OC (Tables 1 and 2), point to indoor sources, such as organic emissions from children and staff, cotton fibers from clothes, wood furniture, crafting in school's activities (with the use of glue), among others (Branis and Sfranek, 2011; Fromme et al., 2008; Kalimeri et al., 2019; Rivas et al., 2014; Weschler & Shields, 1999). OC could also be generated indoor by reactions between oxidants from outdoors and gas-phase compounds emitted indoors, forming secondary organic aerosols (SOA) (Hodas et al., 2016: Long et al., 2000; Wainman et al., 2000; Waring, 2014; Waring & Siegel, 2010, 2013; Waring et al., 2011; Weschler, 2011; Weschler & Shields, 1999). Considering the high indoor OC concentration values reported in this study, especially during period 1 in School 2 and period 2 in School 1 (Table 1), the importance of proper ventilation on the school's environment becomes evident, given the potential toxicity of OC and SOC.

Proper ventilation is known to be an important issue in the school environment. In addition to the emission of particles, the accumulation of fine PM indoors (PM2.5, PM1) can occur in classrooms. Indeed, magnetic grain size distribution measured in School 1 displays a clear difference between indoor and outdoor environments, with a finer magnetic granulometry indoors close to the xARM/SIRM ratio of PM2.5 filters obtained by Leite et al. (2021). Conversely, School 2 samples present a similar magnetic grain size distribution indoor and outdoor typical of the entire PM range (Mitchell & Maher, 2009). School 1 outdoor environment also falls in this latter range. Although both schools have the same architecture, the teacher's interview revealed that all windows of School 2 were changed following the AZF industrial accident of 2001 allowing the windows to be fully open and classrooms properly ventilated, while School 1 has old tilt windows that can never be fully opened, which could have led to the differentiation in size range.

Indoor PM partly results from the penetration of PM from outdoor ambient air via openings (doors and windows) and building cracks (infiltration/exfiltration, Hodas et al., 2016). They can also be transported into the classroom by children and teachers (Chen & Zhao, 2011) via hair, clothes and bags. Here, high PM concentrations inside classrooms are related to poor venting. Indeed, the window ventilation is one of the most important actions to promote good air quality in schools (Almeida et al., 2011; Blondeau et al., 2005; Chithra & Nagendra, 2012, 2014; Rovelli et al., 2014) without automatic venting systems such as here, in Toulouse. The measurements of high  $CO_2$  concentrations in School 1, required by law, also highlight a lack of ventilation especially in the instrumented classroom. A  $CO_2$  concentration of 3,200 ppm is reported in 2018 for this specific classroom (Report of 12/06/2018) and over the 10 classrooms investigated, 7 were above 1,000 ppm. This is due to the aging of the openings and lack of renovation. In the classroom in concern, over the 10 existing windows, only five were still

functioning and only three could be effectively opened by the teacher. The teachers' interviews also indicated that the classroom is ventilated (opening the windows) in the morning during cleaning before the arrival of children and sometimes during the day. As also stated by Mumovic et al. (2009), managing troubles in concentration and the required ventilation is not easy for the teacher especially with absence of official guidelines regarding air pollution except for avoiding outdoor sports during pollution peaks. This study also highlighted that the teachers were not aware of the official  $CO_2$  monitoring in their classrooms. The opening of windows during class time is described as subjecting children to noise, drafts and changes in temperature that have the effect of dispersing the group's attention.

High indoor concentration values of fine PM appear to be the hallmark of the school environment. Indeed, significant PM concentrations are recorded indoors especially in School 1 for period 1 as highlighted by high I/O for PM1 (Table 2) and SIRM values. High concentrations of PM2.5 have also been previously reported indoors in the school environment throughout Europe. A mean I/O ratio of 1.44 has been reported for 122 European schools (Kalimeri et al., 2019). Similar high values were obtained, among others, from 27 schools in Belgium with I/O ratios varying from 1 to 2.7 (Stranger et al., 2008) in 20 schools in Portugal with a mean I/O ratio of 1.45 (Madureira et al., 2012) or from 39 Spanish schools with I/O ratios varying from 1.2 to 1.5 (Rivas et al., 2015).

#### 4.3. Vegetal Passive Collectors to Monitor Airborne Particles and Students-Teacher Involvement

The recorded concentrations in schools from Toulouse are higher than the air quality guidelines (WHO, 2021a) both outside and inside the classrooms. As one of the teachers explicitly stated, teachers and students suspected poor air quality because of the direct view and noise of the large entrance road next to the yard.

Both the informal exchanges between the teachers and researchers and the official interviews led to a co-design of the process and to the co-production of knowledge. Teachers were involved in the design of the intervention by asking the researchers to come more than once into the classrooms and explain the needs of children. From the teacher's point of view and observations, the fact that the sensor was made with almost nothing (verbatim *"trois fois rien"*) and from trees similar to those present in the yard was the predominant factor of endorsement by the children. The fact that children were able to meet the researchers several times, touch and experiment with concrete materials in their classrooms is the most important. Ecoanxiety as defined by Pihkala (2020) could affect all participants (Cunsolo et al., 2020; Dillon & Krasny, 2014; Russell & Oakley, 2016), the children as well as teachers and researchers. Both scientific activities around magnetic phenomena and the use of natural media help the discussion on air quality and its possible impact to be less anxiogenic for all actors.

For the monitoring of air quality in the school environment and the evaluation of children's exposure, the collaboration between the school (teachers, management and students) and the researchers makes it possible to better adapt the points of measurements via the feeling, acceptance and perceptions. Such an approach enables solutions or assessments to emerge. This empowers students and teachers to know their environment and to move toward solutions ranging from conscious acceptance to advocacy, legitimized by collaboration for practical measures. In this case, the co-building of solutions was initiated when preliminary results became available. Notably, they are concerned about the default in ventilation of the classrooms. However, the dialog was abruptly interrupted by the sanitary crisis of 2020.

## 5. Conclusions

A combination of magnetic and chemical methods highlights high ultrafine and fine PM concentrations of airborne particles both inside and outside the classrooms of two elementary schools in the city of Toulouse. Since the main source for PM is identified as traffic, both the proximity to the ring road and a lack of ventilation explain the measured values. Higher outdoor PM1 concentrations in both schools compared to the official traffic stations from downtown Toulouse point toward the fact that the local environment (i.e., distance to a source), here consisting of the proximity to the ring road and a major access road, is of crucial importance for the potential exposures of pupils. In addition, the microenvironments constrained by the architectural and vegetation configuration also have impacts on local PM concentrations. Collaboration between teachers, students, and scientists benefits the knowledge production about the influence of traffic sources in schools, school microenvironments, and environmental education. The use of vegetation as support for particles trapping, in addition to active sampling on filters, benefits both scientific studies by allowing the use of multiple sensors/stations and environmental education by



promoting school children's participation while reducing the anxiety associated with this type of environmental assessment. A comparison of PM accumulation in Toulouse and European schools shows a global tendency for higher PM concentrations in classrooms despite discrepancies in outdoor air quality. Epidemiological surveys inside school environments are urgently required to understand potential health and learning skills impacts on children of PM accumulation inside classrooms and would benefit from passive bio-sensors monitoring.

# **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

# **Data Availability Statement**

Data set for this research is available in Zenodo repositories, under the https://zenodo.org/badge/DOI/10.5281/ zenodo.6362617 (Leite et al., 2022).

### References

- Almeida, S. M., Canha, N., Silva, A., do Carmo Freitas, M., Pegas, P., Alves, C., et al. (2011). Children exposure to atmospheric particles in indoor of Lisbon primary schools. *Atmospheric Environment*, 45(40), 7594–7599. https://doi.org/10.1016/j.atmosenv.2010.11.052
- Annesi-Maesano, I., Baiz, N., Banerjee, S., Rudnai, P., Rive, S., & on behalf of the SINPHONIE Group. (2013). Indoor air quality and sources in schools and related health effects. *Journal of Toxicology and Environmental Health, Part B*, 16(8), 491–550. https://doi.org/10.1080/1093 7404.2013.853609
- Bardelli, F., Cattaruzza, E., Gonella, F., Rampazzo, G., & Valotto, G. (2011). Characterization of road dust collected in Traforo del San Bernardo highway tunnel: Fe and Mn speciation. *Atmospheric Environment*, 45(35), 6459–6468. https://doi.org/10.1016/j.atmosenv.2011.07.035
- Bloemendal, J., King, J. W., Hall, F. R., & Doh, S. J. (1992). Rock magnetism of Late Neogene and Pleistocene deep-sea sediments: Relationship to sediment source, diagenetic processes, and sediment lithology. *Journal of Geophysical Research*, 97(B4), 4361–4375. https://doi. org/10.1029/91JB03068
- Blondeau, P., Iordache, V., Poupard, O., Genin, D., & Allard, F. (2005). Relationship between outdoor and indoor air quality in eight French schools. *Indoor Air*, 15(1), 2–12. 10.1111/j.1600-0668.2004.00263.x
- Braniš, M., & Šafránek, J. (2011). Characterization of coarse particulate matter in school gyms. Environmental Research, 111(4), 485–491. https://doi.org/10.1016/j.envres.2011.03.010
- Castañeda-Miranda, A. G., Böhnel, H. N., Molina-Garza, R. S., & Chaparro, M. A. (2014). Magnetic evaluation of TSP-filters for air quality monitoring. Atmospheric Environment, 96, 163–174. https://doi.org/10.1016/j.atmosenv.2014.07.015
- Chaparro, M. A., Chaparro, M. A., Castaneda-Miranda, A. G., Marie, D. C., Gargiulo, J. D., Lavornia, J. M., et al. (2020). Fine air pollution particles trapped by street tree barks: In situ magnetic biomonitoring. *Environmental Pollution*, 266, 115229. https://doi.org/10.1016/j. envpol.2020.115229
- Chen, C., & Zhao, B. (2011). Review of relationship between indoor and outdoor particles: I/O ratio, infiltration factor and penetration factor. Atmospheric Environment, 45(2), 275–288. https://doi.org/10.1016/j.atmosenv.2010.09.048
- Chiappini, L., Verlhac, S., Aujay, R., Maenhaut, W., Putaud, J. P., Sciare, J., et al. (2014). Clues for a standardised thermal-optical protocol for the assessment of organic and elemental carbon within ambient air particulate matter. *Atmospheric Measurement Techniques*, 7(6), 1649–1661. https://doi.org/10.5194/amt-7-1649-2014
- Chithra, V. S., & Nagendra, S. M. (2014). Characterizing and predicting coarse and fine particulates in classrooms located close to an urban roadway. *Journal of the Air and Waste Management Association*, 64(8), 945–956. https://doi.org/10.1080/10962247.2014.894483
- Chithra, V. S., & Nagendra, S. S. (2012). Indoor air quality investigations in a naturally ventilated school building located close to an urban roadway in Chennai, India. *Building and Environment*, 54, 159–167. https://doi.org/10.1016/j.buildenv.2012.01.016
- Chow, J. C., Chen, L.-W. A., Watson, J. G., Lowenthal, D. H., Magliano, K. A., Turkiewicz, K., & Lehrman, D. E. (2006). PM<sub>2.5</sub> chemical composition and spatiotemporal variability during the California regional PM<sub>10</sub>/PM<sub>2.5</sub> air quality study (CRPAQS). *Journal of Geophysical Research*, 111(D10), D10S04. https://doi.org/10.1029/2005JD006457
- Chow, J. C., Watson, J. G., Fujita, E. M., Lu, Z., Lawson, D. R., & Ashbaugh, L. L. (1994). Temporal and spatial variations of PM<sub>2.5</sub> and PM<sub>10</sub> aerosol in the Southern California air quality study. *Atmospheric Environment*, 28(12), 2061–2080. https://doi.org/10.1016/1352-2310(94)90474-X
- Chow, J. C., Watson, J. G., Kuhns, H., Etyemezian, V., Lowenthal, D. H., Crow, D., et al. (2004). Source profiles for industrial, mobile, and area sources in the Big Bend Regional Aerosol Visibility and Observational Study. *Chemosphere*, 54(2), 185–208. https://doi.org/10.1016/j. chemosphere.2003.07.004
- Chow, J. C., Watson, J. G., Pritchett, L. C., Pierson, W. R., Frazier, C. A., & Purcell, R. G. (1993). The DRI thermal/optical reflectance carbon analysis system: Description, evaluation and applications in U.S. air quality studies. *Atmospheric Environment, Part A: General Topics*, 27(8), 1185–1201. https://doi.org/10.1016/0960-1686(93)90245-T
- Christophoul, F., Regard, V., Martinod, J., & Darrozes, J. (2014). Morphodynamics of the Upper Pleistocene Garonne River (SW France): Conditions of braiding/meandering transition. Retrieved From https://hal.archives-ouvertes.fr/hal-01005191/document
- Cunsolo, A., Harper, S. L., Minor, K., Hayes, K., Williams, K. G., & Howard, C. (2020). Ecological grief and anxiety: The start of a healthy response to climate change? *The Lancet Planetary Health*, 4(7), e261–e263. https://doi.org/10.1016/S2542-5196(20)30144-3
- Dankers, P. H. M. (1978). Magnetic properties of dispersed natural iron-oxides of known grain-size. Ph. D. Thesis, (p.143). University of Utrecht. Dawai, D., Macouin, M., Rousse, S., Leon, J.-F., Gountie Dedzo, M., & Drigo, L. (2021). Tracking airborne pollution with environmental magnetism in a medium-sized African city. Atmosphere, 12(10), 1281. https://doi.org/10.3390/atmos12101281
- Dillon, J., & Krasny, M. E. (2014). Chapter the emotional life of the environmental educator: John Fraser and Carol B. Brandt. In M. E. Krasny & J. Dillon (Eds.), *Trading zones in environmental education* (pp. 133–158). Peter Lang.

#### Acknowledgments

The authors sincerely thank the pedagogical teams and children of Jules Julien and Leo Lagrange schools of Toulouse. The authors acknowledge the support of the city of Toulouse. The authors thank the communication team of the CNRS-délégation Occitanie Ouest for their help along this project. This study was funded by the French National Research Agency ANR-grant number ANR-19-CE04-0008 through the BREATHE project and has received financial support through the NanoEnvi project funded by the CNRS-MITI interdisciplinary programs-Osez l'interdisciplinarité-NanoEnvi. A. d. S. L, was supported by a PhD grant of the French Government and Ministry of Education. This work has been (partially) performed at USPMag lab at Instituto de Astronomia, Geofísica e Ciencias Atmosféricas (IAG) at Universidade de São Paulo (USP) funded by CAPES/ FAPESP/CNPO.



- Dimitroulopoulou, C. (2012). Ventilation in European dwellings: A review. Building and Environment, 47(1), 109–125. https://doi.org/10.1016/j. buildenv.2011.07.016
- Dunlop, D. J. (2002). Theory and application of the Day plot (M<sub>rs</sub>/M<sub>s</sub> versus H<sub>c</sub>/H<sub>c</sub>) 1. Theoretical curves and tests using titanomagnetite data. Journal of Geophysical Research, 107(B3), EPM-4. https://doi.org/10.1029/2001JB000487
- European Environmental Agency. (2021). Air quality in Europe. Web Report no. 15/2021. https://doi.org/10.2800/549289
- Evans, M. E., & Heller, F. (2003). Environmental magnetism: principles and applications of enviromagnetics. Elsevier.
- Ferguson, A., & Solo-Gabriele, H. (2016). Children's exposure to environmental contaminants: An editorial reflection of articles in the IJERPH special issue entitled, "children's exposure to environmental contaminants". *International Journal of Environmental Research and Public Health*, 13(11), 1117. https://doi.org/10.3390/ijerph13111117
- Formenti, P., Caquineau, S., Chevaillier, S., Klaver, A., Desboeufs, K., Rajot, J. L., et al. (2014). Dominance of goethite over hematite in iron oxides of mineral dust from Western Africa: Quantitative partitioning by X-ray absorption spectroscopy. *Journal of Geophysical Research:* Atmospheres, 119(22), 12740–12754. https://doi.org/10.1002/2014JD021668
- Frank, U., & Nowaczyk, N. R. (2008). Mineral magnetic properties of artificial samples systematically mixed from haematite and magnetite. Geophysical Journal International, 175(2), 449–461. https://doi.org/10.1111/j.1365-246X.2008.03821.x
- Fromme, H., Diemer, J., Dietrich, S., Cyrys, J., Heinrich, J., Lang, W., et al. (2008). Chemical and morphological properties of particulate matter (PM<sub>10</sub>, PM<sub>2.5</sub>) in school classrooms and outdoor air. Atmospheric Environment, 42(27), 6597–6605. https://doi.org/10.1016/j. atmosenv.2008.04.047
- Gauderman, W. J., Vora, H., McConnell, R., Berhane, K., Gilliland, F., Thomas, D., et al. (2007). Effect of exposure to traffic on lung development from 10 to 18 years of age: A cohort study. *The Lancet*, 369(9561), 571–577. https://doi.org/10.1016/s0140-6736(07)60037-3
- Gonet, T., & Maher, B. A. (2019). Airborne, vehicle-derived Fe-bearing nanoparticles in the urban environment: A review. Environmental Science & Technology, 53(17), 9970–9991. https://doi.org/10.1021/acs.est.9b01505
- Gonet, T., Maher, B. A., Nyirő-Kósa, I., Pósfai, M., Vaculík, M., & Kukutschová, J. (2021). Size-resolved, quantitative evaluation of the magnetic mineralogy of airborne brake-wear particulate emissions. *Environmental Pollution*, 288, 117808. https://doi.org/10.1016/j.envpol.2021.117808
- Gorka-Kostrubiec, B., & Szczepaniak-Wnuk, I. (2017). Magnetic study of a mixture of magnetite and metallic iron in indoor dust samples. *Air Quality, Atmosphere & Health, 10*(1), 105–116. https://doi.org/10.1007/s11869-016-0412-5
- Górka-Kostrubiec, B., Werner, T., Dytłow, S., Szczepaniak-Wnuk, I., Jeleńska, M., & Hanc-Kuczkowska, A. (2019). Detection of metallic iron in urban dust by using high-temperature measurements supplemented with microscopic observations and Mössbauer spectra. *Journal of Applied Geophysics*, 166, 89–102. https://doi.org/10.1016/j.jappgeo.2019.04.022
- Henderson, S. B., Beckerman, B., Jerrett, M., & Brauer, M. (2007). Application of land use regression to estimate long-term concentrations of traffic-related nitrogen oxides and fine particulate matter. *Environmental Science & Technology*, 41(7), 2422–2428. https://doi.org/10.1021/ es0606780
- Hodas, N., Loh, M., Shin, H. M., Li, D., Bennett, D., McKone, T. E., et al. (2016). Indoor inhalation intake fractions of fine particulate matter: Review of influencing factors. *Indoor Air*, 26(6), 836–856. https://doi.org/10.1111/ina.12268
- Hofman, J., & Samson, R. (2014). Biomagnetic monitoring as a validation tool for local air quality models: A case study for an urban street canyon. Environment International, 70, 50–61. https://doi.org/10.1016/j.envint.2014.05.007
- Hofman, J., Stokkaer, I., Snauwaert, L., & Samson, R. (2013). Spatial distribution assessment of particulate matter in an urban street canyon using biomagnetic leaf monitoring of tree crown deposited particles. *Environmental Pollution*, 183, 123–132. https://doi.org/10.1016/j. envpol.2012.09.015
- Höfner, S., & Schütze, A. (2021). Air quality measurements and education: Improving environmental awareness of high school students. Frontiers in Sensors, 2(April), 1–12. https://doi.org/10.3389/fsens.2021.657920
- Jelenska, M., Gorka-Kostrubiec, B., Werner, T., Kadzialko-Hofmokl, M., Szczepaniak-Wnuk, I., Gonet, T., & Szwarczewski, P. (2017). Evaluation of indoor/outdoor urban air pollution by magnetic, chemical and microscopic studies. *Atmospheric Pollution Research*, 8(4), 754–766. https://doi.org/10.1016/j.apr.2017.01.006
- Jordanova, N., Jordanova, D., Henry, B., Le Goff, M., Dimov, D., & Tsacheva, T. (2006). Magnetism of cigarette ashes. Journal of Magnetism and Magnetic Materials, 301(1), 50–66. https://doi.org/10.1016/j.jmmm.2005.06.008
- Kalimeri, K. K., Bartzis, J. G., Sakellaris, I. A., & de Oliveira Fernandes, E. (2019). Investigation of the PM<sub>2.5</sub>, NO<sub>2</sub> and O<sub>3</sub> I/O ratios for office and school microenvironments. *Environmental Research*, 179, 108791. https://doi.org/10.1016/j.envres.2019.108791
- Kelepertzis, E., Argyraki, A., Botsou, F., Aidona, E., Abel, S., & Szabo, C. (2019). Tracking the occurrence of anthropogenic magnetic particles and potentially toxic elements (PTEs) in house dust using magnetic and geochemical analyses. *Environmental Pollution*, 245, 909–920. https:// doi.org/10.1016/j.envpol.2018.11.072
- Kelly, F. J., & Fussell, J. C. (2015). Air pollution and public health: Emerging hazards and improved understanding of risk. Environmental Geochemistry and Health, 37(4), 631–649. https://doi.org/10.1007/s10653-015-9720-1
- Khreis, H., Kelly, C., Tate, J., Parslow, R., Lucas, K., & Nieuwenhuijsen, M. (2017). Exposure to traffic-related air pollution and risk of development of childhood asthma: A systematic review and meta-analysis. *Environment International*, 100, 1–31. https://doi.org/10.1016/j. envint.2016.11.012
- Kim, J. J., Smorodinsky, S., Lipsett, M., Singer, B. C., Hodgson, A. T., & Ostro, B. (2004). Traffic-related air pollution near busy roads: The East bay children's respiratory health study. American Journal of Respiratory and Critical Care Medicine, 170(5), 520–526. https://doi. org/10.1164/rccm.200403-2810C
- King, J. W., & Channell, J. E. (1991). Sedimentary magnetism, environmental magnetism, and magnetostratigraphy. *Reviews of Geophysics*, 29(S1), 358–370. https://doi.org/10.1002/rog.1991.29.s1.358
- Leite, A. D. S., Léon, J.-F., Macouin, M., Rousse, S., Trindade, R. I. F. D., Proietti, A., et al. (2021). PM<sub>2.5</sub> magnetic properties in relation to urban combustion sources in southern west Africa. *Atmosphere*, 12(4), 496. https://doi.org/10.3390/atmos12040496
- Leite, A. D. S., Mendonça, C. A., Moraes, P. L., & Ustra, A. T. (2018). A procedure for quantitative characterization of superparamagnetic minerals in environmental magnetism. *Geophysical Journal International*, 215(3), 1974–1984. https://doi.org/10.1093/gji/ggy395
  - Leite, A. D. S., Rousse, S., Léon, J. -F., Trindade, I. F. R., Haoues-Jouve, S., Carvallo, C., Macouin, M., et al. (2022). Dataset–Barking up the right tree: Using tree bark to track airborne particles in school environment and link science to society. Zenodo Repository. https://doi.org/10.5281/ zenodo.6362617
  - Liati, A., Pandurangi, S. S., Boulouchos, K., Schreiber, D., & Arroyo Rojas Dasilva, Y. (2015). Metal nanoparticles in diesel exhaust derived by in-cylinder melting of detached engine fragments. Atmospheric Environment, 101, 34–40. https://doi.org/10.1016/j.atmosenv.2014.11.014
  - Limo, J., Paturi, P., & Makinen, J. (2018). Magnetic biomonitoring with moss bags to assess stop-and-go traffic induced particulate matter and heavy metal concentrations. Atmospheric Environment, 195, 187–195. https://doi.org/10.1016/j.atmosenv.2018.09.062

- Long, C. M., Suh, H. H., & Koutrakis, P. (2000). Characterization of indoor particle sources using continuous mass and size monitors. Journal of the Air & Waste Management Association, 50(7), 1236–1250. https://doi.org/10.1080/10473289.2000.10464154
- Madureira, J., Paciência, I., & De Oliveira Fernandes, E. (2012). Levels and indoor-outdoor relationships of size-specific particulate matter in naturally ventilated Portuguese schools. *Journal of Toxicology and Environmental Health Part A: Current Issues*, 75(22–23), 1423–1436. https://doi.org/10.1080/15287394.2012.721177
- Maher, B. A. (1988). Magnetic properties of some synthetic sub-micron magnetites. *Geophysical Journal International*, 94(1), 83–96. https://doi.org/10.1111/j.1365-246X.1988.tb03429.x
- Marié, D. C., Chaparro, M. A., Lavornia, J. M., Sinito, A. M., Castaneda Miranda, A. G., Gargiulo, J. D., & Bohnel, H. N. (2018). Atmospheric pollution assessed by in situ measurement of magnetic susceptibility on lichens. *Ecological Indicators*, 95(August), 831–840. https://doi. org/10.1016/j.ecolind.2018.08.029
- Martins, V., Faria, T., Diapouli, E., Manousakas, M. I., Eleftheriadis, K., Viana, M., & Almeida, S. M. (2020). Relationship between indoor and outdoor size-fractionated particulate matter in urban microenvironments: Levels, chemical composition and sources. *Environmental Research*, 183(February), 109203. https://doi.org/10.1016/j.envres.2020.109203
- Mendell, M. J., & Heath, G. A. (2005). Do indoor pollutants and thermal conditions in schools influence student performance? A critical review of the literature. *Indoor Air*, 15(1), 27–52. https://doi.org/10.1111/j.1600-0668.2004.00320.x
- Mitchell, R., & Maher, B. A. (2009). Evaluation and application of biomagnetic monitoring of traffic-derived particulate pollution. Atmospheric Environment, 43(13), 2095–2103. https://doi.org/10.1016/j.atmosenv.2009.01.042
- Moreno, E., Sagnotti, L., Dinares-Turell, J., Winkler, A., & Cascella, A. (2003). Biomonitoring of traffic air pollution in Rome using magnetic properties of tree leaves. Atmospheric Environment, 37(21), 2967–2977. https://doi.org/10.1016/S1352-2310(03)00244-9
- Muñoz, D., Aguilar, B., Fuentealba, R., & Prendez, M. (2017). Environmental studies in two communes of Santiago de Chile by the analysis of magnetic properties of particulate matter deposited on leaves of roadside trees. Atmospheric Environment, 152, 617–627. https://doi. org/10.1016/j.atmosenv.2016.12.047
- Mumovic, D., Palmer, J., Davies, M., Orme, M., Ridley, I., Oreszczyn, T., et al. (2009). Winter indoor air quality, thermal comfort and acoustic performance of newly built secondary schools in England. *Building and Environment*, 44(7), 1466–1477. https://doi.org/10.1016/j. buildenv.2008.06.014
- Nakazato, R. K., Esposito, M. P., Cardoso-Gustavson, P., Bulbovas, P., Pedroso, A. N. V., de Assis, P. I. L. S., & Domingos, M. (2018). Efficiency of biomonitoring methods applying tropical bioindicator plants for assessing the phytoxicity of the air pollutants in SE, Brazil. *Environmental Science and Pollution Research*, 25(20), 19323–19337. https://doi.org/10.1007/s11356-018-2294-6
- Ouafo-Leumbe, M.-R., Galy-Lacaux, C., Liousse, C., Pont, V., Akpo, A., Doumbia, T., et al. (2018). Chemical composition and sources of atmospheric aerosols at Djougou (Benin). *Meteorology and Atmospheric Physics*, 130(5), 591–609. https://doi.org/10.1007/s00703-017-0538-5
- Paraskevopoulou, D., Liakakou, E., Gerasopoulos, E., Theodosi, C., & Mihalopoulos, N. (2014). Long-term characterization of organic and elemental carbon in the PM<sub>2.5</sub> fraction: The case of Athens, Greece. Atmospheric Chemistry and Physics, 14(23), 13313–13325. https://doi. org/10.5194/acp-14-13313-2014
- Parviainen, A., Papaslioti, E. M., Casares-Porcel, M., & Garrido, C. J. (2020). Antimony as a tracer of non-exhaust traffic emissions in air pollution in Granada(S Spain) using lichen bioindicators. *Environmental Pollution*, 263, 114482. https://doi.org/10.1016/j.envpol.2020.114482 Pihkala, P. (2020). Eco-anxiety and environmental education. *Sustainability*, 12(23), 10149. https://doi.org/10.3390/su122310149
- Pirela, S. V., Sotiriou, G. A., Bello, D., Shafer, M., Lee, K., Castranova, V., et al. (2015). Consumer exposures to laser printer-emitted engineered nanoparticles: A case study of life-cycle implications from nano-enabled products. *Enabled Products*, 9(6), 760–768. https://doi.org/10.3109 /17435390.2014.976602
- Pope, C. A., III, & Dockery, D. W. (2006). Health effects of fine particulate air pollution: Lines that connect. Journal of the Air & Waste Management Association, 56(6), 709–742. https://doi.org/10.1080/10473289.2006.10464485
- Remoundaki, E., Kassomenos, P., Mantas, E., Mihalopoulos, N., & Tsezos, M. (2013). Composition and mass closure of PM<sub>2.5</sub> in urban environment (Athens, Greece). Aerosol and Air Quality Research, 13(1), 72–82. https://doi.org/10.4209/aaqr.2012.03.0054
- Rice, D., & Barone, S., Jr. (2000). Critical periods of vulnerability for the developing nervous system: Evidence from humans and animal models. Environmental Health Perspectives, 108(suppl 3), 511–533. https://doi.org/10.1289/ehp.00108s3511
- Rivas, I., Viana, M., Moreno, T., Bouso, L., Pandolfi, M., Alvarez-Pedrerol, M., et al. (2015). Outdoor infiltration and indoor contribution of UFP and BC, OC, secondary inorganic ions and metals in PM<sub>2.5</sub> in schools. *Atmospheric Environment*, 106, 129–138. https://doi.org/10.1016/j. atmosenv.2015.01.055
- Rivas, I., Viana, M., Moreno, T., Pandolfi, M., Amato, F., Reche, C., et al. (2014). Child exposure to indoor and outdoor air pollutants in schools in Barcelona, Spain. *Environment International*, 69, 200–212. https://doi.org/10.1016/j.envint.2014.04.009
- Rovelli, S., Cattaneo, A., Nuzzi, C. P., Spinazzè, A., Piazza, S., Carrer, P., & Cavallo, D. M. (2014). Airborne particulate matter in school classrooms of northern Italy. *International Journal of Environmental Research and Public Health*, 11(2), 1398–1421. https://doi.org/10.3390/ ijerph110201398
- Russell, C., & Oakley, J. (2016). Engaging the emotional dimensions of environmental education. *Canadian Journal of Environmental Education*, 21, 13–22.
- Sagnotti, L., Macri, P., Egli, R., & Mondinio, M. (2006). Magnetic properties of atmospheric particulate matter from automatic air sampler stations in Latium (Italy): Toward a definition of magnetic fingerprints for natural and anthropogenic PM<sub>10</sub> sources. *Journal of Geophysical Research*, 111(12), 1–17. https://doi.org/10.1029/2006JB004508
- Sagnotti, L., Taddeucci, J., Winkler, A., & Cavallo, A. (2009). Compositional, morphological, and hysteresis characterization of magnetic airborne particulate matter in Rome, Italy. *Geochemistry, Geophysics, Geosystems*, 10(8). Q08Z06. https://doi.org/10.1029/2009GC002563
- Sagnotti, L., & Winkler, A. (2012). On the magnetic characterization and quantification of the superparamagnetic fraction of traffic-related urban air-borne PM in Rome, Italy. Atmospheric Environment, 59, 131–140. https://doi.org/10.1016/j.atmosenv.2012.04.058
- Shi, Y., Ji, Y., Sun, H., Hui, F., Hu, J., Wu, Y., et al. (2015). Nanoscale characterization of PM<sub>2.5</sub> airborne pollutants reveals high adhesiveness and aggregation capability of soot particles. *Scientific Reports*, 5(April), 1–11. https://doi.org/10.1038/srep11232
- Stranger, M., Potgieter-Vermaak, S. S., & Van Grieken, R. (2008). Characterization of indoor air quality in primary schools in Antwerp, Belgium. *Indoor Air*, 18(6), 454–463. https://doi.org/10.1111/j.1600-0668.2008.00545.x
- Szabados, M., Csákó, Z., Kotlík, B., Kazmarová, H., Kozajda, A., Jutraz, A., et al. (2021). Indoor air quality and the associated health risk in primary school buildings in Central Europe–The InAirQ study. *Indoor Air*, *31*(4), 989–1003. https://doi.org/10.1111/ina.12802
- Szöyi, M., Sagnotti, L., & Hirt, A. M. (2007). On leaf magnetic homogeneity in particulate matter biomonitoring studies. *Geophysical Research Letters*, 34(6), L06306. https://doi.org/10.1029/2006GL029076
- Tham, K. W. (2016). Indoor air quality and its effects on humans—A review of challenges and developments in the last 30 years. *Energy and Buildings*, 130, 637–650. https://doi.org/10.1016/j.enbuild.2016.08.071

- Urbat, M., Lehndorff, E., & Schwark, L. (2004). Biomonitoring of air quality in the cologne conurbation using pine needles as a passive sampler— Part I: Magnetic properties. Atmospheric Environment, 38(23), 3781–3792. https://doi.org/10.1016/j.atmosenv.2004.03.061
- Vezzola, L. C., Muttoni, G., Merlini, M., Rotiroti, N., Pagliardini, L., Hirt, A. M., & Pelfini, M. (2017). Investigating distribution patterns of airborne magnetic grains trapped in tree barks in Milan, Italy: Insights for pollution mitigation strategies. *Geophysical Journal International*, 210(2), 989–1000. https://doi.org/10.1093/gji/ggx232
- Viana, M., Maenhaut, W., ten Brink, H. M., Chi, X., Weijers, E., Querol, X., et al. (2007). Comparative analysis of organic and elemental carbon concentrations in carbonaceous aerosols in three European cities. *Atmospheric Environment*, 41(28), 5972–5983. https://doi.org/10.1016/j. atmosenv.2007.03.035
- Vornanen-Winqvist, C., Jarvi, K., Andersson, M. A., Duchaine, C., Letourneau, V., Kedves, O., et al. (2020). Exposure to indoor air contaminants in school buildings with and without reported indoor air quality problems. *Environment International*, 141(October 2019), 105781. https://doi. org/10.1016/j.envint.2020.105781
- Wainman, T., Zhang, J., Weschler, C. J., & Lioy, P. J. (2000). Ozone and limonene in indoor air: A source of submicron particle exposure. Environmental Health Perspectives, 108(12), 1139–1145. https://doi.org/10.1289/ehp.001081139
- Waring, M. S. (2014). Secondary organic aerosol in residences: Predicting its fraction of fine particle mass and determinants of formation strength. *Indoor Air*, 24(4), 376–389. https://doi.org/10.1111/ina.12092
- Waring, M. S., & Siegel, J. A. (2010). The influence of HVAC systems on indoor secondary organic aerosol formation. ASHRAE Transactions, 116(1), 556–571.
- Waring, M. S., & Siegel, J. A. (2013). Indoor secondary organic aerosol formation initiated from reactions between ozone and surface-sorbed D-limonene. *Environmental science & technology*, 47(12), 6341–6348. https://doi.org/10.1021/es400846d
- Waring, M. S., Wells, J. R., & Siegel, J. A. (2011). Secondary organic aerosol formation from ozone reactions with single terpenoids and terpenoid mixtures. Atmospheric Environment, 45(25), 4235–4242. https://doi.org/10.1016/j.atmosenv.2011.05.001
- Weschler, C. (2011). Chemistry in indoor environments: 20 years of research. Indoor Air, 21(3), 205–218. https://doi.org/10.1111/j.1600-0668.2011.00713.x
- Weschler, C. J., & Shields, H. C. (1999). Indoor ozone/terpene reactions as a source of indoor particles. Atmospheric Environment, 33(15), 2301–2312. https://doi.org/10.1016/S1352-2310(99)00083-7
- West, S. E., Büker, P., Ashmore, M., Njoroge, G., Welden, N., Muhoza, C., et al. (2020). Particulate matter pollution in an informal settlement in Nairobi: Using citizen science to make the invisible visible. *Applied Geography*, 114, 102133. https://doi.org/10.1016/j.apgeog.2019.102133
- WHO. (2021a). Global air quality guidelines: Particulate matter (PM<sub>2.5</sub> and PM<sub>10</sub>), ozone, nitrogen dioxide, sulfur dioxide and carbon monoxide. World Health Organization.
- WHO. (2021b). Literature review on chemical pollutants in indoor air in public settings for children and overview of their health effects with a focus on schools, kindergartens and day-care centres: Supplementary publication to the screening tool for assessment of health. World Health Organization.
- Winkler, A., Amoroso, A., Di Giosa, A., & Marchegiani, G. (2021). The effect of Covid-19 lockdown on airborne particulate matter in Rome, Italy: A magnetic point of view. *Environmental Pollution*, 291, 118191. https://doi.org/10.1016/j.envpol.2021.118191
- Xu, H., Guinot, B., Shen, Z., Ho, K. F., Niu, X., Xiao, S., et al. (2015). Characteristics of organic and elemental carbon in PM<sub>2.5</sub> and PM<sub>0.25</sub> in indoor and outdoor environments of a middle school: Secondary formation of organic carbon and sources identification. *Atmosphere*, 6(3), 361–379. https://doi.org/10.3390/atmos6030361
- Zhang, C., Huang, B., Piper, J. D., & Luo, R. (2008). Biomonitoring of atmospheric particulate matter using magnetic properties of Salix matsudana tree ring cores. Science of the Total Environment, 393(1), 177–190. https://doi.org/10.1016/j.scitotenv.2007.12.032
- Zmirou, D., Gauvin, S., Pin, I., Momas, I., Just, J., Sahraoui, F., et al. (2002). Five epidemiological studies on transport and asthma: Objectives, design and descriptive results. *Journal of Exposure Science and Environmental Epidemiology*, 12(3), 186–196. https://doi.org/10.1038/sj/ jea/7500217

# **References From the Supporting Information**

- Barmparesos, N., Assimakopoulos, M. N., Assimakopoulos, V. D., Loumos, N., Sotiriou, M. A., & Koukoumtzis, A. (2018). Indoor air quality and thermal conditions in a primary school with a green roof system. *Atmosphere*, 9(2), 75. https://doi.org/10.3390/atmos9020075
- Becerra, J. A., Lizana, J., Gil, M., Barrios-Padura, A., Blondeau, P., & Chacartegui, R. (2020). Identification of potential indoor air pollutants in schools. *Journal of Cleaner Production*, 242, 118420. https://doi.org/10.1016/j.jclepro.2019.118420
- Braniš, M., Šafránek, J., & Hytychová, A. (2009). Exposure of children to airborne particulate matter of different size fractions during indoor physical education at school. *Building and Environment*, 44(6), 1246–1252. https://doi.org/10.1016/j.buildenv.2008.09.010
- Diapouli, E., Chaloulakou, A., Mihalopoulos, N., & Spyrellis, N. (2008). Indoor and outdoor PM mass and number concentrations at schools in the Athens area. *Environmental Monitoring and Assessment*, 136(1), 13–20. https://doi.org/10.1007/s10661-007-9724-0
- Di Gilio, A., Farella, G., Marzocca, A., Giua, R., Assennato, G., Tutino, M., & De Gennaro, G. (2017). Indoor/outdoor air quality assessment at school near the steel plant in Taranto (Italy). Advances in Meteorology, 1–7. https://doi.org/10.1155/2017/1526209
- Fromme, H., Dietrich, S., Twardella, D., Heitmann, D., Schierl, R., Kiranoglu, M., & Liebl, B. (2006). Indoor air concentrations of particulate matter (PM<sub>10</sub> and PM<sub>2.5</sub>) in German schools. WIT Transactions on Ecology and the Environment, 86, 393–399. https://doi.org/10.2495/ AIR06039
- Fromme, H., Twardella, D., Dietrich, S., Heitmann, D., Schierl, R., Liebl, B., & Rüden, H. (2007). Particulate matter in the indoor air of classrooms—Exploratory results from Munich and surrounding area. *Atmospheric Environment*, 41(4), 854–866. https://doi.org/10.1016/j. atmosenv.2006.08.053
- Janssen, N. A., van Vliet, P. H., Aarts, F., Harssema, H., & Brunekreef, B. (2001). Assessment of exposure to traffic related air pollution of children attending schools near motorways. Atmospheric Environment, 35(22), 3875–3884. https://doi.org/10.1016/S1352-2310(01)00144-3
- Kovačević, R., Tasic, V., Živković, M., Živković, N., Dordevic, A., Manojlović, D. D., & Jovasevic-Stojanovic, M. (2015). Mass concentrations and indoor-outdoor relationships of PM in selected educational buildings in Nis, Serbia. *Chemical Industry and Chemical Engineering Quar*terly/CICEQ, 21(1), 149–157. https://doi.org/10.2298/CICEQ140207013K
- Pacitto, A., Stabile, L., Viana, M., Scungio, M., Reche, C., Querol, X., et al. (2018). Particle-related exposure, dose and lung cancer risk of primary school children in two European countries. *Science of the Total Environment*, 616, 720–729. https://doi.org/10.1016/j.scitoteny.2017.10.256
- Šafránek, J., Turčová, I., Braniš, M., & Hájek, M. (2017). Exposure of children to aerosol during PE lessons. *Envigogika*, 12(2). https://doi.org/10.14712/18023061.554



- Tofful, L., & Perrino, C. (2015). Chemical composition of indoor and outdoor  $PM_{2.5}$  in three schools in the city of Rome. *Atmosphere*, 6(10), 1422–1443. https://doi.org/10.3390/atmos6101422
- van Drooge, B. L., Rivas, I., Querol, X., Sunyer, J., & Grimalt, J. O. (2020). Organic air quality markers of indoor and outdoor PM<sub>25</sub> aerosols in primary schools from Barcelona. International Journal of Environmental Research and Public Health, 17(10), 3685. https://doi.org/10.3390/ ijerph17103685
- Viana, M., Rivas, I., Querol, X., Alastuey, A., Alvarez-Pedrerol, M., Bouso, L., et al. (2015). Partitioning of trace elements and metals between quasi-ultrafine, accumulation and coarse aerosols in indoor and outdoor air in schools. *Atmospheric Environment*, 106, 392–401. https://doi. org/10.1016/j.atmosenv.2014.07.027
- Wichmann, J., Lind, T., Nilsson, M. M., & Bellander, T. (2010). PM<sub>2.5</sub>, soot and NO<sub>2</sub> indoor–outdoor relationships at homes, pre-schools and schools in Stockholm, Sweden. Atmospheric Environment, 44(36), 4536–4544. https://doi.org/10.1016/j.atmosenv.2010.08.02